Author's response to reviews

Title: The effectiveness of community engagement in public health interventions for disadvantaged groups: A meta-analysis

Authors:

    Alison O'Mara-Eves (a.omara-eves@ioe.ac.uk)
    Ginny Brunton (g.brunton@ioe.ac.uk)
    Josephine Kavanagh (j.kavanagh@ioe.ac.uk)
    Sandy Oliver (s.oliver@ioe.ac.uk)
    Farah Jamal (f.jamal@uel.ac.uk)
    James Thomas (j.thomas@ioe.ac.uk)

Version: 4 Date: 15 December 2014

Author's response to reviews:

(please see note to editors below)